Author's response to reviews

Title: Posterior Leukoencephalopathy following Repair of an Ileo-cecal Anastomosis Breakdown: A Case Report and Review of the Literature

Authors:

Pascal O. Zinn (pzinn@bidmc.harvard.edu)
Rivka R. Colen (RRCOLEN@partners.org)
Ekkehard M. Kasper (ekasper@bidmc.harvard.edu)
Clark C. Chen (Clark.Chen@dfci.harvard.edu)

Version: 2 Date: 12 July 2010

Author's response to reviews:

MS: 1066972118396216

Case report
Posterior Leukoencephalopathy following Repair of an Ileo-cecal Anastomosis Breakdown: A Case Report and Review of the Literature
Pascal O. Zinn, Rivka R. Colen, Ekkehard M. Kasper and Clark C. Chen
Journal of Medical Case Reports

Dear Dr Zinn,

Thank you for submitting your manuscript (above) to Journal of Medical Case Reports. There are a number of formatting changes required before your manuscript can be considered for peer review. These are listed below, but we advise that you also refer to our instructions for authors http://jmedicalcasereports.com/info/instructions/.

Attached is a template explaining exactly how a manuscript must be structured. Please refer to this template and also address the following specific points:

Please structure your abstract into the following three sections:

Introduction An introduction about why this case is important and needs to be reported. Please include information on whether this is the first report of this kind in the literature.

Case presentation Brief details of what the patient presented with, including the patient’s age, sex and ethnic background.

Conclusion A brief conclusion of what the reader should learn from the case report and what the clinical impact will be. Is it an original case report of interest to a particular clinical speciality of medicine or will it have a broader clinical impact across medicine? Please include information on how it will significantly advance our knowledge of a particular disease etiology or drug mechanism.
Please include the ethnicity of the patient in the case presentation section of the manuscript.

In keeping with journal style, please could you split your conclusion section into a discussion (providing any additional relevant information which puts the case in context) and a conclusion (stating what can be learnt from the case).

Please could you confirm that you have obtained written informed consent from the patient for publication of the manuscript and figures used. A statement to this effect must appear in a consent section at the end of the manuscript, before the reference list. It is a requirement to have written informed patient consent from the patient for publication of the case and we will not proceed with your manuscript until you have confirmed this.